Author’s response to reviews

Title: The effect of nursing participation in the design of a critical care information system: a case study in a Chinese hospital

Authors:

Yanhong Qin (qinyanhong1023@sina.com)
Ranyun Zhou (zhoury307yyhlb@163.com)
Qiong Wu (hkhl2yz@163.com)
Xiaodi Huang (hxdpunk@163.com)
Xinli Chen (572469084@qq.com)
Weiwei Wang (1013855013@qq.com)
Xun Wang (1260996156@qq.com)
Hua Xu (1697907561@qq.com)
Jing Zheng (studentzj0921@163.com)
Siyu Qian (sq992@uowmail.edu.au)
Changqing Bai (mlp1604@sina.com)
Ping Yu (ping@uow.edu.au)

Version: 4 Date: 16 Nov 2017

Author’s response to reviews:

1 - Please include figure legends after the references for the figures
Response: figure legends have been listed after the references. See page 29.

2 - Please ensure that there is only one unique set of initials for each author in authors' contributions section. We believe one author is represented twice in the section; XW.
Response: this has been corrected.
3 - Please remove the response to reviewer letter from the file inventory, as it is no longer needed at this stage of the editorial process.

Response: removed.

4 - At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response: the revised manuscript is a clean version. Figures were uploaded as separate files.